Artificial Intelligence 2B 2004–2005 PROVISIONAL TIMETABLE

AI2B consists of two Modules, each consisting of two Tasks:

Module 3 (Mark Steedman)

Task 5: Learning from data. 9 lectures.

Task 6: Coping with incomplete knowledge. 9 lectures.

Module 4 (Alan Bundy)

Task 7: Coping with a changing world. 9 lectures.

Task 8: Dealing with other agents. 5 lectures.

Lecture and tutorial schedule

Lectures are at 4:10 pm in Lecture Theatre 3, Appleton Tower (Mondays and Thursdays) and Room G8, 1 George Square (Tuesdays).

Week w/c Monday Tuesday Thursday Tutorial topic 1 10 Jan Task 5 Task 5 Task 5 – 2 17 Jan Task 5 Task 5 Task 5 Task 5 3 24 Jan Task 5 Task 5 Task 5 Task 5 4 31 Jan Task 6 Task 6 Task 6 Task 5 5 7 Feb Task 6 Task 6 Task 6 Task 6 6 14 Feb Task 6 Task 6 Task 6 Task 6 7 21 Feb Task 7 Task 7 Task 7 Task 7 8 28 Feb Task 7 Task 7 Task 7 Task 7 9 7 Mar Task 7 Task 7 Task 7 Task 7 10 14 Mar Task 8 Task 8 Task 8 Task 8 11 21 Mar Task 8 Task 8 – Task 8

Practical schedule

Practical Set by Submission deadline B1 Mark Steedman Friday 28 Jan (Week 3) B2 Mark Steedman Friday 18 Feb (Week 6) B3 Alan Bundy Friday 11 Mar (Week 9) B4 Alan Bundy Friday 25 Mar (Week 11)

Submission deadlines for practicals will be at 4pm on Fridays.